\documentclass[11pt]{article}
\usepackage{amssymb,graphicx,enumerate} 
\newcommand{\ncmd}{\newcommand}

\ncmd{\R}{\mathbb{R}} 
\ncmd{\C}{\mathbb{C}}
\ncmd{\Z}{\mathbb{Z}}
\ncmd{\N}{\mathbb{N}}
\ncmd{\F}{\mathbb{F}}
\newtheorem{thm}{Theorem}
\newtheorem{defn}[thm]{Definition}
\newtheorem{question}{Question}
\newtheorem{remark}[thm]{Remark}
\ncmd{\proposer}[1]{\par proposed by~\emph{#1}.\par\medskip}

\author{Frank Morgan}
\author{Pierre Pansu}


\begin{document}
\begin{thebibliography}{99}
 \bibitem{AdSa}
 {\sc I. M. Adelstein and M.R. Sandoval,} 
 {\it The G-invariant spectrum and non-orbifold singularities,}
 Arch. Math., 109 (2017), 563-573.
 \bibitem{Sutton}
 {\sc C. Sutton,} 
{\it Equivariant isospectrality and Sunada's method,}
Arch. Math., 95 (2010), 75-85.
\end{thebibliography}



\section*{A list of open problems in Differential Geometry}

Good open problems play an indispensable role in the development of differential geometry. All speakers and participants of the conference Modern Trends in Differential Geometry (S\~ao Paulo, July 2018) were invited to include open problems. We thank the contributors and the organizers. We think that this collection reflects many facets of geometry, its deep roots, and its profusion of fruit.

\hfill Frank Morgan, Pierre Pansu

\section{Can you hear an orbifold singularity?}
\proposer{Ian Adelstein, Yale}

Mark Kac in the American Mathematical Monthly in 1966 famously asked if one can hear the shape of a drum, by which he meant whether the Laplace spectrum of a manifold determines its isometry class. John Milnor quickly answered this question in the negative, by producing a pair of isospectral yet non-isometric $16$ dimensional tori. Many subsequent mathematicians have wondered which geometric properties are audible, i.e. determined by the spectrum of the Laplacian. 

An important open question is 
\begin{question}
Can one hear the presence of an orbifold singularity, i.e. whether or not there exists a pair of isospectral orbifolds, one of which has singular points whereas the other does not and is therefore a manifold. 
\end{question}
The paper \cite{Sutton} discusses this question. 

Recent work has shown that non-orbifold singularities are inaudible to the G-invariant spectrum. In \cite{AdSa} the authors construct a pair of orbit spaces with equivalent G-invariant spectra, yet where one space is isometric to an orbifold whereas the other admits a non-orbifold singularity.




\section{Riemannian manifolds with curvature bounds}
\proposer{Lashi Bandara, Universit\"at Potsdam, Germany}

Can a complete metric with Ricci curvature bounded below be $L^\infty$ approximated by one with two-sided Ricci curvature bounds?

Here is a precise statement.
\begin{question}
For every $\ell,k>0$, there exist $C,L,K>0$ with the following effect. Let $(M,g)$ be a complete Riemannian manifold with injectivity radius $inj(M,g)\geq \ell$ and Ricci curvature $Ric(g)\geq k$. Then there exists a metric $h$ on $M$ with $inj(M,h)\geq L$ and $|Ric(h)|\leq K$ such that
$$
\frac{1}{C}\,g\leq h\leq C\,g.
$$
\end{question}

Let me motivate this problem. During my thesis, my supervisor Alan
McIntosh and I studied a problem called the Kato square root problem on
manifolds. In the Euclidean context, this was a 40 year outstanding
problem, which Alan was involved in for that entire period of time. We
considered this question in a geometric setting. In our work, we prove that this 
problem can be solved for metrics that have Ricci bounded above and below, 
as well as injectivity radius, \cite{BandaraMcIntosh}. In a
later paper, \cite{Bandara}, I showed that the solution is stable under $L^\infty$
perturbations as expressed in the question. 

\begin{thebibliography}{99}
\bibitem{BandaraMcIntosh}
{\sc L. Bandara, A. McIntosh},
{\it The Kato square root problem on vector bundles with generalised bounded geometry}, 
J. Geom. Anal. {\bf 26} (2016), no. 1, 428--462.

\bibitem{Bandara}
{\sc L. Bandara},
{\it Continuity of solutions to space-varying pointwise linear elliptic equations},
 Publ. Mat. {\bf 61} (2017), no. 1, 239--258.
\end{thebibliography}


\section{Reducibility of the holonomy of flat manifolds}
\proposer{Renato Bettiol, City University of New York (Lehman College)}

\begin{question}
Give an alternative, geometric proof that the holonomy representation of a closed flat manifold is reducible.
\end{question}

By the Bieberbach Theorem, a closed flat $n$-dimensional manifold is finitely covered by a flat $n$-torus. The group of deck transformations is identified with the manifold's holonomy group $H$, which is a finite subgroup of $O(n)$. In a very algebraic paper, \cite{GHAS}, Hiss and Szczepanski proved that the (orthogonal) action of $H$ in $\R^n$ is always reducible, i.e., $\R^n$ admits a decomposition as the orthogonal direct sum of nontrivial $H$-invariant subspaces. Although this has important geometric consequences (e.g., it implies that all closed flat manifolds admit non-homothetic flat deformations, see \cite{BDP}), the proof does not seem to provide much more geometric insight and is completely algebraic (even relies on the classification of finite simple groups). It is reasonable to expect this fact should have a purely geometric proof, and that this could shed some light on important questions about flat manifolds.

\begin{thebibliography}{99}
\bibitem{BDP}
{\sc Bettiol, R. G., Derdzinski, A. and Piccione, P.},
{\it Teichm\"uller theory and collapse of flat manifolds}, 
Ann. Mat. Pura Appl. {\bf 197}, (2018), no. 4, 1247--1268.
\bibitem{charlap}
{\sc L. S. Charlap},
{\it Bieberbach Groups and Flat Manifolds},
Universitext, Springer, NewYork (1986). 
\bibitem{GHAS}
{\sc G. Hiss, A. Szczepanski},
{\it On torsion free crystallographic groups},
J. Pure Appl. Algebra {\bf 74}, (1991) 39--56. 
\end{thebibliography}

\section{Biorthogonal curvature}
\proposer{Renato Bettiol, City University of New York (Lehman College)}

\begin{question}
Does $S^2\times T^2$ admit a Riemannian metric with positive biorthogonal curvature?
\end{question}

The biorthogonal curvature of a $4$-manifold is defined, for each tangent plane $\sigma$, as the average of the sectional curvature of $\sigma$ and $\sigma^\perp$. Positive biorthogonal curvature is an intermediate condition between positive sectional curvature and positive scalar curvature. Clearly, $S^2 \times T^2$ admits metrics with nonnegative biorthogonal curvature, but it is unknown if they can be deformed into metrics with (strictly) positive biorthogonal curvature. Settling this matter would be a crucial step in classifying non-simply connected closed $4$-manifolds with positive biorthogonal curvature (the simply-connected classification, up to homeomorphisms, was obtained in \cite{AsianBettiol}).

\begin{thebibliography}{99}
\bibitem{AsianBettiol}
{\sc R. G. Bettiol}, 
{\it Four-dimensional manifolds with positive biorthogonal curvature},
Asian J. Math {\bf 21} (2017), no. 2, 391--396.

\bibitem{PAMSBettiol}
{\sc R. G. Bettiol}, 
{\it Positive biorthogonal curvature on $S^2 \times S^2$.} 
Proc. Amer. Math. Soc. {\bf 142} (2014), no. 12, 4341--4353.

\end{thebibliography}

\section{Branch points of area-minimizing surfaces}
\proposer{Camillo De Lellis, Institute for Advanced Study, Princeton}

$\Sigma$ is a smooth, complete $(m+n)$-dimensional Riemannian manifold, $\Gamma \subset \Sigma$ a smooth $(m-1)$-dimensional oriented submanifold and $T$ an area minimizing integer rectifiable current which spans $\Gamma$. 
We are interested in the boundary regularity of $T$, more precisely on the size of ${\rm Reg}_b (T)\subset \Gamma$, the (relatively open) set of boundary regular points, and its complement ${\rm Sing}_b (T)= \Gamma\setminus {\rm Reg}_b (T)$. The question is pertinent only in codimension $n\geq 2$, because the celebrated work of Hardt and Simon \cite{HS} shows that ${\rm Sing}_b (T) =\emptyset$ when $n=1$. 

In \cite{DDHM} we show that ${\rm Reg}_b (T)$ is dense in $\Gamma$. We also give an example of a $2$-dimensional $T$ with a sequence $\{P_k\}\subset {\rm Sing}_b (T)$ of crossing singularities accumulating towards a branching singularity $P\in {\rm Sing}_b (T)$. This in sharp contrast with the interior regularity theory, because we know that interior singular points of area-minimizing $2$-dimensional $T$ are isolated  (cf. \cite{Chang,DSS1,DSS2}).

\begin{question}
1. Does ${\rm Sing}_b (T)$ have zero $(m-1)$-dimensional Hausdorff measure? 

2. If yes, does ${\rm Sing}_b (T)$ have (Hausdorff) dimension at most $m-2$? 

3. A related question is: can we find an example as in \cite{DDHM} where the sequence of accumulating singularities $\{P_k\}\subset {\rm Sing}_b (T)$ are of branching type? If yes, then one could imagine a Cantor type construction giving a negative answer to 2.

4. White in \cite{White} conjectures that if a $2$-dimensional area-minimizing currents spans a real analytic closed curve, then it has finitely many (interior and boundary) singularities and hence finite topological type. 

5. The example in \cite{DDHM} is topologically a disk. Is it possible to give an example of a smooth closed curve $\Gamma \subset \mathbb R^{2+n}$ which bounds an area-minimizing $2$-dimensional current with infinite topology?  
\end{question}

\begin{thebibliography}{99}

\bibitem{Chang}
{\sc S. X. Chang},
{\it Two-dimensional area minimizint integral currents are classical minimal surfaces}.
J. Amer. Math. Soc. {\bf 1} (1988), no. 4, 699--778.

\bibitem{DSS1}
{\sc C. De Lellis, E. Spadaro, L. Spolaor},
{\it Regularity theory for $2$-dimensional almost minimal currents II: branched center manifold}.
Ann. PDE {\bf 3} (2017), no. 2, Art. 18, 85 pp.

\bibitem{DSS2}
{\sc C. De Lellis, E. Spadaro, L. Spolaor},
{\it Regularity theory for $2$-dimensional almost minimal currents II: blow up}.
arXiv:1508.05510 To appear in Jour. Diff. Geom.


\bibitem{DDHM}
{\sc C. De Lellis, G. De Philippis, J. Hirsch, A. Massaccesi}, 
{\it On the boundary behavior of mass-minimizing integral currents}.
arXiv:1809.09457

\bibitem{HS}
{\sc R. Hardt, L. Simon},
{\it Boundary regularity and embedded solutions of the oriented Plateau problem}.
Ann. of Math. (2) {\bf 110} (1979), no. 3, 439--386.

\bibitem{White}
{\sc B. White},
{\it Classical area minimizing surfaces with real-analytic boundaries}.
Acta Math. {\bf 179} (1997), no. 2, 295--305. 

\end{thebibliography}


\section{Manifolds modelled on flag manifolds}
\proposer{Elisha Falbel, Sorbonne Universit\'e, Paris}

Flag manifolds are homogeneous manifolds obtained as quotients of a complex group
by a Borel subgroup.  One of the simplest examples is the complex full flag manifold of complex dimension three $F_{12}=SL(3,\C)/B$, where $B$ is the subgroup of upper triangular matrices.  Each of the real forms $SL(3,\R)$ and $SU(2,1)$ have  a unique compact orbit which is a model for  {\it flat path geometry} structures and  
{\it spherical CR structures} respectively.  Both orbits have real dimension three.  Here is natural question, certainly very difficult as a variant of uniformization problems, open even in dimension three:
   
\begin{question}
Which manifolds can be modeled on an orbit of a real form in a space of flags? 
\end{question}

Back to real dimension three,  one can define flag structures on a real 3-manifold by a choice of two complex lines in its complexified tangent space.  A very special case appears when
the manifold admits a totally real  immersion in $F_{12}$.  The complex lines arises when one considers  natural foliations in $F_{12}$ intersected with the real immersion.  Because it is so special, this flag structure can be thought as a flat
structure even though it is not locally equivalent to one of the real orbits.  This situation leads to the following
 
\begin{question}
What is the homotopy classification of totally real immersions of real $3$-manifolds in the complex full flag manifold $F_{12}$? 
\end{question}

Gromov's $H$-principle applies here, so the problem should belong to homotopy theory.

\begin{thebibliography}{99}
\bibitem{FV}
{\sc E. Falbel, J. Veloso}, 
{\it Flag structures on real 3-manifolds}
arXiv:1804.11096
\end{thebibliography}

\section{Area minimizing projective spaces in the projective space with the Berger metric}
\proposer{Olga Gil-Medrano, University of Valencia, Spain.}

A classical result due to M. Berger  \cite{B} and A. T. Fomenko \cite{Fo} is that for the real projective $n$-dimensional space with the standard metric, for all $0<k<n$, the projective spaces obtained by projection of the $k$-dimensional great spheres (that are totally geodesic submanifolds) are the only $k$-dimensional volume minimizing in their homology class.\par

\bigskip
A Berger sphere is the Riemannian manifold $(S^{2n+1}, g_{\mu} )$ where the metric $g_{\mu}$ for $\mu > 0$ is defined as follows:
 we denote by $J$ the complex structure of $\R^{2n+2}\simeq \C^{n+1}$ and by $H$ the Hopf vector field $JN,$ where $N(p) =p$ is the outwards unit normal of $S^{2n+1}$,  then
$$g_{\mu}(H,H)=\mu g(H,H)=\mu,\quad (g_{\mu})|_{H^{\perp}}=g|_{H^{\perp}},\quad \mbox{ and }\quad  g_{\mu}(H,H^{\perp})=0,$$
where $\perp$ means orthogonal with respect to the standard metric $g$. By passing to the quotient we can construct the corresponding Berger metrics on the projective spaces $( \R P^{2n+1}, g_\mu)$
\par
\bigskip
In \cite{B-GM} it is shown that for $( \R P^3,g_\mu)$  the projections of the equatorial spheres are not totally geodesic surfaces, except for the round sphere $\mu=1$,  but they are always minimal surfaces and in \cite{GM} it is shown that in fact \emph{area minimizing projective planes in} $(\R P^3, g_\mu)$ \emph{are exactly the equatorial projective planes.} 

The key point for the proof is the existence of area minimizing projective planes in $3$-manifolds stablished in \cite{B-B-E-N}.\par

\bigskip

\begin{question} For ${2n+1}>3$ and $0<k<{2n+1}$, are projective subspaces obtained by projection of the $k$-dimensional equatorial spheres minimal submanifolds of the Berger projective space $( \R P^{2n+1}, g_\mu)$? Are they the only $k$-dimensional volume minimizing cycle in their homology classes?
\end{question}

\begin{remark}
In \cite{B-GM} it has been shown that Berger projective space of dimension $3$ can be realized as the manifold $(T^rS^2, g^S)$ of vectors of norm $r$ on the unit sphere; more precisely $(T^rS^2, g^S)$ is isometric to $( \R P^3,4g_{r^2})$. The isometry preserves the $S^1$-bundle structure of both manifolds, namely the one determined by the fibers of the Hopf fibration in the projective space and the natural one in the tangent space.This isometry is specific of dimension $3$.
\end{remark} 


\begin{thebibliography}{99}

 \bibitem{B} 
{\sc M. Berger}, 
{\it Du c\^ot\'e de chez Pu},
Ann. Scient. Ec. Norm. Sup. \textbf{5}, (1972), 1--44.

\bibitem{B-GM} 
{\sc V. Borrelli and O. Gil-Medrano}, 
{\it Area-minimizing vector fields on $2$-spheres}, 
J. Reine angew. Math. \textbf{640}, (2010), 85--99.

\bibitem{B-B-E-N} 
{\sc H. Bray, S Brendle, M. Eichmair and A. Neves}, 
{\it Area-minimizing Projective Planes in $3$-Manifolds}, 
Comm. Pure and Appl. Math. Vol. \textbf{LXIII}, (2010), 1237--1247.

\bibitem{Fo} 
{\sc A. T. Fomenko}, 
{\it Minimal compacta in Riemannian manifolds and Reifenberg's conjecture}, 
Math. USSR, \textbf{6}, (1972), 1037--1066. 

\bibitem{GM}
{\sc O. Gil-Medrano}, 
{\it Area minimizing projective planes on the projective space of dimension $3$ with the Berger metric.}, 
C. R. Acad. Sci. Paris, Ser. I, \textbf{354}, (2016), 415--417

\end{thebibliography}




\section{Bi-invariant metrics and multiplicity of conjugate points}

\proposer{Claudio Gorodski, University of S\~ao Paulo, Brazil}

Let $G$ be a compact connected Lie group equipped with a bi-invariant
Riemannian metric. Then $G$ is a symmetric space, namely,
the Riemann curvature tensor~$R$ is parallel. It follows that
the Jacobi equation along a geodesic $\gamma$, given by
\[ -J'' +R(X,J)X=0, \]
where $X=\gamma'$, has constant coefficients in an
orthonormal parallel frame along $\gamma$.
More explicitly, the Jacobi operator is
\[ R(X,\cdot)X=\frac14\mathrm{ad}_X^2. \]
Here $\mathrm{ad}_X$ is a skew-symmetric operator,
so the non-zero eigenvalues of the Jacobi operator
occur in pairs and hence all conjugate points are of even order;
equivalently, the index of any geodesic segment is even (Bott~\cite{Bott}). 

We ask whether this property characterizes bi-invariant metrics
on Lie groups among the left-invariant ones (of course, it can be asked
among all Riemannian metrics, but we would like to consider the
algebraic structure of Lie groups). 

\begin{question} 
  Assume that a left-invariant Riemannian metric is given
  on a compact connected Lie group
  $G$ such that the index of any geodesic segment is even.
  Must the metric be bi-invariant?
\end{question}

It is not difficult to see that for a given Riemannian metric on $G$, 
the index of any geodesic segment
is even if and only if, for any
two non-conjugate points $p$, $q\in G$, the 
energy functional
\[ E(\gamma)=\frac12\int|\dot\gamma|^2\,dt \]
defined on the Hilbert manifold of $H^1$-curves joining $p$ and $q$
is a perfect Morse function. Complete Riemannian metrics with the latter
property were called \emph{pointwise taut} by Terng and Thorbergsson
in~\cite{TT}. 


\begin{thebibliography}{99}
\bibitem{Bott}
{\sc R. Bott,}
{\it An application of the Morse theory to the topology of Lie-groups},
Bull.\ Soc.\ Math.\ France \textbf{84} (1956), 251--281.

\bibitem{TT}
{\sc C.-L. Terng and G. Thorbergsson,} 
{\it Taut immersions into complete Riemannian manifolds},
Tight and taut submanifolds (Berkeley, CA, 1994), 181--228,
Math.\ Sci.\ Res.\ Inst.\ Publ.\, 32, Cambridge Univ.\ Press,
Cambridge, 1997. 
\end{thebibliography}


\section{Toral manifolds and positive scalar curvature}
\proposer{Bernhard Hanke, Augsburg}

The classification of closed manifolds admitting Riemannian metrics of positive scalar curvature has been a major research topic for decades. 
While for simply connected manifolds of dimension at least $5$ a complete classification has been achieved by Gromov-Lawson \cite{GL} and Stolz \cite{St}, the picture remains largely unclear even for finite fundamental groups.

\begin{question}[\cite{Ros}, Conjecture 1.2]  \label{conj:ros} Let $M$ be a connected  closed manifold with finite fundamental group of odd order. 
Assume that the universal cover of $M$ admits a metric of positive scalar curvature. Does $M$ admit a metric of positive scalar curvature? 
\end{question} 
% a corriger : admits -> admit

\begin{remark} \label{rem:Ros} Question \ref{conj:ros} has a positive answer for $\dim M  \leq 2$, and for $\dim M = 3$ it follows from the geometrisation theorem. 
For $\dim M = 4$, counterexamples were constructed in \cite{HKW}. 
For $\dim M \geq 5$ Question \ref{conj:ros} is open. 
\end{remark} 
% a corriger : Conjecture -> Question, holds -> has a positive answer

\begin{defn} \label{atoral}
Let $M^d$ be a connected closed $d$-manifold, and let $p$ be a prime. 
The manifold $M$ is called {\em $p$-toral}, if there exist cohomology classes  $c_1, \ldots, c_d \in H^1(M ;  \F_p )$ with $c_1 \cup \cdots \cup c_d  \neq 0 \in H^d(M ; \F_p)$. 
Otherwise $M$ is called {\em $p$-atoral}. 
\end{defn} 

\begin{remark} \label{ex:atoral}  
\begin{enumerate}[(i)] 
\item  \label{torus_bordism} The $d$-torus $T^d = (S^1 \times \cdots \times S^1)^d$ is $p$-toral for all $p$, and so are all closed manifolds which are oriented bordant to $T^d$ over the classifying space $B (\Z/p)^d$. 
\item    The real projective spaces $\R P^{2m+1}$ are $2$-toral for all $m$. 
\item   \label{toral_orient}  If $p$ is odd and $M$ is $p$-toral, then $M$ is orientable. 
\item    If  $p$ is odd and  $\dim M > \mathrm{rank} \big( H^1 ( B \pi_1(M) ; \F_p) \big)$, then $M$ is $p$-atoral.  
This follows from the fact that the classifying map $M \to B \pi_1(M)$ induces an isomorphism in $H^1(- ; \F_p)$ and for odd $p$ each element in  $H^1(M ; \F_p)$ has square $0$. 
%a corriger : B \pi_1(M) -> M
\item  In \cite{Ha15} we give a positive answer to Question \ref{conj:ros}, if $\dim M \geq 5$, $\pi_1(M)$ is an elementary abelian $p$-group for some odd $p$ and $M$ is $p$-atoral. 
\end{enumerate}
\end{remark} 
% a corriger : positive


\begin{question} \label{conj:main} Let $M$ be a connected closed manifold admitting a metric of positive scalar curvature. Does this imply that $M$ is $p$-atoral for all odd  $p$? 
\end{question} 
% a corriger : Does this imply

The answer to Question \ref{conj:main} is positive if $\dim M  \leq 2$. 
For $\dim M = 3$ we recall \cite[Theorem 8.1]{GL-IHES} that the prime decomposition of $M$ (which we can assume to be orientable by Remark \ref{ex:atoral} (iii)) only contains copies of $S^1 \times S^2$ and manifolds with universal covers homotopy equivalent to $S^3$. 
This implies that for odd $p$ all prime summands of $M$, and hence $M$ itself, are $p$-atoral. 



Let $p$ be odd and let $M^d$ be a $p$-toral non-spin manifold with finite fundamental group of odd order. 
Such manifolds can be constructed along Remark \ref{ex:atoral} (i) for all $d \geq 4$. 
The universal cover of $M$ is a closed simply connected non-spin manifold, and therefore it admits a metric of positive scalar curvature  in case $d \geq 5$ by \cite[Corollary C]{GL}.
Together with Remark \ref{rem:Ros}, we conclude that if Question \ref{conj:main} has a positive answer, then Question  \ref{conj:ros} has a negative answer for $\dim M \geq 4$. 

\begin{thebibliography}{99}
\bibitem{GL}
{\sc M. Gromov, H. B. Lawson, Jr.}, 
{\it The classification of simply connected manifolds of positive scalar curvature}, 
Ann. of Math. (2) {\bf 111} (1980), no. 3, 423--434. 
\bibitem{GL-IHES}
{\sc M. Gromov, H. B. Lawson, Jr.},
{\it Positive scalar curvature and the {D}irac operator on complete
              {R}iemannian manifolds},
Inst. Hautes \'{E}tudes Sci. Publ. Math. {\bf 58} (1983), 83--196.
\bibitem{Ha15}
{\sc B. Hanke},
{\it Bordism of elementary abelian groups via inessential {B}rown-{P}eterson homology},
J. Topol. {\bf 9} (2016), no. 3, 725--746.
\bibitem{HKW}
{\sc B. Hanke, D. Kotschick, J. Wehrheim},
{\it Dissolving four-manifolds and positive scalar curvature},
Math. Z. {\bf 245} (2003), no. 3, 545--555.
 \bibitem{Ros}
{\sc J. Rosenberg},
{\it {$C^\ast$}-algebras, positive scalar curvature, and the {N}ovikov conjecture. {III}},
Topology {\bf 25} (1986), no. 3, 319--336.
\bibitem{St}
{\sc S. Stolz}, 
{\it Simply connected manifolds of positive scalar curvature}, 
Ann. of Math. (2) {\bf 136} (1992), no. 3, 511--540. 
\end{thebibliography}


\section{Coarse embeddings}
\proposer{Dominique Hulin and Pierre Pansu, Paris-Sud (Paris-Saclay)}

A map between metric spaces is a coarse embedding if there exist functions $\omega$ and $\Omega$, where $\omega$ tends to $+\infty$ with its argument, such that for every $D>0$, pairs of points at distance $D$ are mapped to pairs at distance between $\omega(D)$ and $\Omega(D)$. This large class arose in convex geometry in the 1960's, and later in geometric group theory, where coarse embeddability of a fundamental group into Hilbert space turns out to make it rather tame from the topological viewpoint. Therefore the wider problem of coarse embeddability of certain classes of metric spaces into certain classes of Banach spaces has received a lot of attention, see references in \cite{EMN}.

For nonlinear targets, large scale analogues of topological dimension have been studied. They are nondecreasing under coarse embeddings, see \cite{GBAD}. They take integer values. Only recently did noninteger valued invariants arise, like variants of the conformal dimension of the ideal boundary of a hyperbolic metric space, see \cite{PP,HMKT}, or sharp metric cotype, see \cite{EMN}.

\begin{question}
Find more numerical invariants of metric spaces that are nondecreasing under coarse embeddings.
\end{question}

A remarkable fact about coarse embeddings is that they can sometimes be made harmonic, see \cite{YBDH}.

\begin{question}
Find applications of the harmonic map approximation of coarse embeddings.
\end{question}

\begin{thebibliography}{99}
\bibitem{GBAD}
{\sc G. Bell, A. Dranishnikov}, 
{\it Asymptotic dimension}, 
Topology Appl. {\bf 155} (2008), no. 12, 1265--1296. 
\bibitem{YBDH}
{\sc Y. Benoist, D. Hulin},
{\it Harmonic quasi-isometric maps II : negatively curved manifolds},
arXiv:1702.04369
\bibitem{EMN}
{\sc A. Eskenazis, M. Mendel, A. Naor},
{\it Nonpositive curvature is not coarsely universal},
arXiv:1808.02179
\bibitem{HMKT}
{\sc D. Hume, J. Mackay, R. Tessera},
{\it Poincar\'e profiles of groups and spaces},
arXiv:1707.02151
 \bibitem{PP}
{\sc P. Pansu},
{\it Large scale conformal maps},
arXiv:1604.01195
\end{thebibliography}

\section{Ricci pinching on solvable Lie groups}
\proposer{Jorge Lauret, C\'ordoba, Argentina}

\begin{question}
For solvable Lie groups $G$, show that solvsolitons are the only local maxima of the Ricci pinching functional $g\mapsto F(g)=\frac{Scal(g)^2}{|Ric(g)|^2}$ on left-invariant Riemannian metrics on $G$.
\end{question}

A solvsoliton is an invariant Riemannian metric $g$ such that $Ric(g)=cI+D$ where $D$ is a derivation of the Lie algebra $\mathfrak{g}$.

\begin{thebibliography}{99}
\bibitem{Lauret}
{\sc J. Lauret, C.E. Will}, 
{\it The Ricci pinching functional on solvmanifolds}, 
arXiv:1808.01380
\end{thebibliography}

\section{Classification problems and Poisson structures}
\proposer{Rui Loja Fernandes, University of Illinois}

R. Bryant (see the Appendix in \cite{Bryant01}) observed that certain classification problems in geometry, which may be called of ``finite type'', can be formulated as the problem of integrating a Lie algebroid $A\to X$ to a Lie groupoid $\mathcal{G}\rightrightarrows X$ (Lie's III Theorem). Examples of such classification problems include:
\begin{enumerate}
\item[(i)] Metrics of constant sectional curvature.
\item[(ii)] Metrics of Hessian type \cite{Bryant14}.
\item[(iii)] Bochner-K\"ahler metrics \cite{Bryant01}.
\item[(iv)] Symplectic connections with special holonomy \cite{CMS96,CS09}.
\end{enumerate}
The Lie groupoid integrating the corresponding Lie algebroid gives a presentation of the stack of the moduli problem in question (see, e.g., \cite{FS14})

On the other hand, given a Poisson manifold $(M,\pi)$, it is well known that its cotangent bundle $A=T^*M$ has a natural Lie algebroid structure. If this Lie algebroid is integrable, then the source 1-connected integration $\mathcal{G}\rightrightarrows X$ is a symplectic groupoid: the space of arrows carries a multiplicative symplectic form $\omega\in\Omega^2(\mathcal{G})$.

It turns out that in all the problems above (and in others of finite type), the Lie algebroid $A$ relevant for the classification problem arises as the cotangent Lie algebroid of some Poisson manifold. However, in some of these problems, there is no natural symplectic or Poisson structure present, and when there is one, it does not seem to arise from the symplectic structure $\omega$ on the Lie groupoid integrating the Lie algebroid $A$. Still, the corresponding moduli space has a ``symplectic nature''.

\begin{question}
Explain the existence and the role of the symplectic nature of the groupoid/algebroid and its relevance for the geometry of the moduli spaces of geometric structures of finite type.
\end{question}

\begin{thebibliography}{99}

\bibitem{Bryant01}
{\sc R. Bryant}, 
{\it Bochner-{K}{\"a}hler Metrics},
J. of Amer. Math. Soc. {\bf 14} (2001), no. 3,  623-715.

\bibitem{Bryant14}
{\sc R. Bryant}, 
{\it Notes on exterior differential systems}, 
Preprint arXiv:1405.3116.

\bibitem{CMS96}
{\sc Q.-S. Chi, S. Merkulov and L. Schwachh{\"o}fer},
{\it On the existence of infinite series of exotic holonomies},
Invent. Math., {\bf 126} (1996), no.2,  391--411.


\bibitem{CS09}
{\sc M. Cahen and L. Schwachh\"{o}fer},
{\it Special symplectic connections},
 J. Differential Geom. {\bf 83} (2009), no. 2,  229--271.


\bibitem{FS14}
{\sc R.L. Fernandes and I. Struchiner},
{\it The classifying {L}ie algebroid of a geometric structure {I}: {C}lasses of coframes},
Trans. Amer. Math. Soc. {\bf 366} (2014), no. 5,  2419--2462.

\end{thebibliography}


\section{Morse index of embedded minimal surfaces}

\proposer{Davi Maximo, University of Pennsylvania}

\begin{question}
\begin{itemize}
  \item Do there exist embedded minimal surfaces with finite genus and Morse index $4$ ?

  \item More focussed: in the $1$-parameter deformation of Costa's surface, does the index stay constant?
\end{itemize}
\end{question}

Complete embedded minimal surfaces of index $3$ have been recently ruled out in \cite{CM}.

\begin{thebibliography}{99}
\bibitem{CM}
{O. Chodosh, D. Maximo},
{\it On the topology and index of minimal surfaces II},
arXiv:1808.06572
 
\end{thebibliography}


\section{On the Hodge spectra of lens spaces}
\proposer{Roberto Miatello, C\'ordoba, Argentina}

\begin{question}
- Construct congruence lattices which are norm$_1$  and norm$_1*$- isospectral
in all dimensions (see \cite{LMR}).

- Are there families of $p$-isospectral lens spaces for all $p$, with more than two elements?

- Give a procedure to go from $p$-isospectral lens spaces (for all $p$) of dimension $m$ to dimension $m + 1$.

- Establish connections with toric geometry (see \cite{MH}).
\end{question}

\begin{thebibliography}{99}

\bibitem{LMR}
{\sc E. Lauret, R. Miatello and J.-P. Rossetti}, 
{\it Spectra of lens spaces from $1$-norm spectra of congruence lattices}, 
IMRN (2016), n$^0$4, 1054--1089.

\bibitem{MH}
{\sc H. Mohades, B. Honari}, 
{\it On a relation between spectral theory of lens spaces and Ehrhart theory}, 
Indag. Math. (N.S.) {\bf 28} (2017), no. 2, 556--565.
\end{thebibliography}



\section{Isoperimetric Problem in $\C P^2$}
\proposer{Frank Morgan, Williams College, Williamstown}

\begin{question}
Prove that geodesic spheres provide the least-perimeter way to enclose prescribed volume in $CP^2$. 
\end{question}
There is enough symmetry to deduce that they have constant mean curvature but not enough to prove them optimal by symmetry arguments.

\begin{thebibliography}{99}
\bibitem{Berger}
{\sc M. Berger,} 
{\it A Panoramic View of Riemannian Geometry,} 
Springer-Verlag, 2003, p. 318.
\end{thebibliography}

\section{Triple Bubble in $\R^3$}
\proposer{Frank Morgan, Williams College, Williamstown}


\begin{question}
Prove that the pictured standard triple soap bubble is the least-perimeter way to enclose and separate three given volumes in $\R^3$. 
\end{question}


\begin{figure}
\begin{center}
\includegraphics[width=3in]{triple_bubble.jpg}
\caption{The triple bubble}
\break
Image by John M. Sullivan \break http://torus.math.uiuc.edu/jms/Images/ \break by permission, all rights reserved.
\end{center}
\end{figure}

The corresponding Double Bubble Theorem was proved in $\R^3$ by Hutchings, Morgan, Ritor\'e and Ros in 2002 and in $\R^n$ by Ben Reichardt in 2008. The double and triple bubble theorems in $\R^n$ with Gaussian density were announced in 2018 by Milman and Neeman.

\begin{thebibliography}{99}
\bibitem{MorganGMT}
{\sc Frank Morgan}, 
{\it Geometric Measure Theory: a Beginner's Guide,} 
Academic Press, 5th edition, 2016, section 13.2.
\end{thebibliography}

\section{Homogeneous Riemannian manifolds with nontrivial nullity}
\proposer{Carlos Olmos, C\'ordoba, Argentina}

\begin{question}
1. If the normal holonomy group of an irreducible and full homogeneous submanifold $M^n$ of the sphere with $n \geq 2$ does not act transitively, then $M$ is the orbit of an $s$-representation.

2. The index of an irreducible symmetric space which is different from $G_2/SO(4)$ (or its symmetric dual) coincides with its reflective index,\cite{BO}.

3. Open questions about homogeneous Riemannian manifolds with nontrivial nullity. 
\begin{itemize}
  \item Are there examples which are not topologically trivial?
  \item Are there examples $M = G/H$, with $G$ non-solvable?
  \item Are there K\"ahler examples?
  \item Are there examples in any dimension $d \geq 5$?
\end{itemize}
\end{question}


\begin{thebibliography}{99}
 \bibitem{BO} 
{\sc J. Berndt, C. Olmos},
{\it Maximal totally geodesic submanifolds and index of symmetric spaces}, 
J. Differen. Geom. \textbf{104} (2016), no. 2, 187--217.
 \bibitem{DSOV} 
{\sc A. J. Di Scala, C. E. Olmos, F. Vittone},
{\it Homogeneous Riemannian manifolds with non-trivial nullity},
arXiv:1802.02642
\end{thebibliography}

\section{Constant mean curvature in homogeneous $3$-manifolds}

\proposer{Joaqu\'\i n P\'erez, University of Granada, Spain}

Here are questions on constant mean curvature (CMC) spheres which arose from the study of the isoperimetric problem in $3$-dimensional, simply-connected, noncompact homogeneous manifolds $X$ diffeomorphic to $\R^3$. 

\begin{question}
\begin{itemize}
  \item Do CMC spheres about a point $x$ in such a space form a foliation of $X-\{x\}$? 
  \item Could this be a way of proving embeddedness of CMC spheres in general? 
\end{itemize}
\end{question}


It is well known \cite{MMPR} that CMC spheres in $X$ exist exactly for values of the mean curvature $H$ lying in an interval of the form $(H(X),\infty)$ where $H(X)$ is called the critical mean curvature of $X$. Furthermore, spheres of CMC $H$ are unique up to congruencies, when they exist. This uniqueness leads to a well-defined notion of center of symmetry for CMC spheres. Finally, it is also known that if we take any point $x$ in $X$, the class of CMC spheres with center $x$ is a real analytic $1$-parameter manifold. It is not known (but expected) if CMC spheres are embedded in general (they are known to be Alexandrov embedded).

I include two classical problems which I partly surveyed in my talk:

\begin{question}
\begin{itemize}
  \item Calabi-Yau problem. For an embedded minimal surface in $\R^3$, does complete imply proper ?
  \item Hoffman-Meeks conjecture. For a complete embedded minimal surface of genus $g$ and $k$ ends and finite total curvature, $k\leq g+2$.
\end{itemize}
\end{question}



\begin{thebibliography}{99}

 \bibitem{MMPR} 
{\sc W. H. Meeks III, P. Mira, J. P\'erez, A. Ros,} 
{\it Constant mean curvature spheres in homogeneous three-manifolds,}
arXiv:1706.09394

\end{thebibliography}


\section{Spherical submetries}
\proposer{Marco Radeschi, Notre Dame}

A submetry is a map with equidistant fibers. In \cite{ALMR}, we show that every submetry from the round sphere is defined by an algebra of polynomials which is both maximal and Laplacian (i.e. stable under the Laplacian). The converse is true: every maximal Laplacian algebra of polynomials defines a spherical submetry.

\begin{question}\label{qradeschi}
Is every Laplacian algebra of polynomials maximal?
\end{question}

When the submetry is given by a global quotient $S^{n-1}\to S^{n-1}/G$
for some representation $\rho:G\subset O(n)$, the Laplacian algebra is simply
the algebra of invariant polynomials. In this context, the question would
solve the following problem in Classical Invariant Theory: can one compute
the invariant polynomials of $\rho\oplus\rho : G\to O(2n)$ in terms of the invariant
polynomials of polynomials of $\rho$? If Question \ref{qradeschi} was true then, letting $x_1,\ldots,x_n,y_1,\ldots,y_n$ be the coordinates of $\R^n\oplus\R^n$, it would follow that the
algebra of invariants of $\rho\oplus\rho$ is the Laplacian algebra generated by the
invariant polynomials of $\rho$ and $\delta(x_1,\ldots,y_n)=\sum_i x_i y_i$. In \cite{RMMR}, we showed that the answer to the question above is yes, when
the Laplacian algebra is generated by polynomials of degree $2$. Even
to prove this rather basic case, though, we had to generalize Weyl's First
Fundamental Theorem of Invariant theory for $O(n)$, to the more general
setup of submetries.

\begin{thebibliography}{99}
\bibitem{ALMR}
{\sc A. Lytchak, M. Radeschi}, 
{\it Algebraic nature of singular Riemannian foliations in spheres}, 
J. Reine Ang. Math. (Crelle) {\bf 744} (2018), 265--273.
DOI: https://doi.org/10.1515/crelle-2016-0010

\bibitem{RMMR}
{\sc R. Mendes, M. Radeschi}, 
{\it Singular Riemannian foliations and their quadratic basic polynomials}, 
arXiv:1611.02067
\end{thebibliography}


\section{Minimax minimal surfaces}
\proposer{Tristan Rivi\`ere, ETH Z\"urich}

We consider the space of Sobolev $W^{3,2}$-immersions of a given closed surface $\Sigma^2$ into a closed riemannian manifold $N^n$ of dimension $n$ greater than or equal to $3$. This space is denoted by $W^{3,2}_{imm}(\Sigma^2,N^n)$.
 An admissible family in this space is a subset of the power set  ${\mathcal P}(W^{3,2}_{imm}(\Sigma^2,N^n))$ which is invariant under the action of any homeomorphism of $W^{3,2}_{imm}(\Sigma^2,N^n)$ isotopic to the identity. 
Let  ${\mathcal A}$ be an admissible family with non zero minimax value (i.e. non zero width)
\[
\beta_{\mathcal A}:=\inf_{A\in {\mathcal A}}\sup_{\Phi\in A} \mbox{Area}(\Phi)>0 .
\]
We prove that there exists a smooth surface $S$ satisfying 
\[
genus(S)\le genus(\Sigma) ,
\]
 and a smooth, possibly branched, minimal immersion $\Phi_{\mathcal A}$ of  $S$ into $N^n$ such that 
\[
\beta_{\mathcal A}=\mbox{Area}(\Phi_{\mathcal A}) .
\]
The admissible family is called  ``homotopic of dimension $d$'' if it is given by continuous mappings from the $d$-dimensional sphere $S^d$ into  $W^{3,2}_{imm}(\Sigma^2,N^n)$. We prove that for such a family the Morse index of the minimal branched immersion is bounded by $d$ : 
\[
\mbox{Index}(\Phi_{\mathcal A})\le d .
\]

\begin{question}
Prove the lower bound
\[
d\le \mbox{Index}(\Phi_{\mathcal A})+\mbox{Null}(\Phi_{\mathcal A}) ,
\]
where $\mbox{Null}(\Phi_{\mathcal A})$ is the {\it nullity} of $\Phi_{\mathcal A}$ that is the dimension of the space of its Jacobi fields.
\end{question}

\begin{question}
Prove that there exists infinitely many distinct minimal branched 2-dimensional immersions in $N^n$.
\end{question}

\begin{thebibliography}{99}
\bibitem{TR2018}
{\sc T. Rivi\`ere},
{\it Lower Semi-Continuity of the Index in the Viscosity Method for Minimal Surfaces},
arXiv:1808.00426, to appear in Int. Math. Res. Not.
\bibitem{APTR2018}
{\sc A. Pigati, T. Rivi\`ere},
{\it A proof of the multiplicity one conjecture for min-max minimal surfaces in arbitrary codimension},
arXiv:1807.04205
\bibitem{APTR2017}
{\sc A. Pigati, T. Rivi\`ere},
{\it The regularity of parametrized integer stationary varifolds in two dimensions},
arXiv:1708.02211
\bibitem{TRIHES}
{\sc T. Rivi\`ere}, 
{\it A viscosity method in the min-max theory of minimal surfaces}, 
Publ. Math. Inst. Hautes \'Etudes Sci. {\bf 126} (2017), 177--246.

\end{thebibliography}


\section{Gromov-Hausdorff convergence of K\"ahler Ricci flow}
\proposer{Gang Tian, BICMR, Peking University}

Let $X$ be a compact K\"ahler $n$-manifold with $K_X\geq 0$ and Kodaira dimension $\kappa=n$. Then the normalized Ricci flow converges to a K\"ahler-Einstein current, smoothly on the ample locus (where $K_X>0$). J. Song proved that the diameter stays bounded. Therefore the completion of the smooth part is a metric on the canonical model of $X$. This does not suffice to imply Gromov-Hausdorff convergence, which has been proved only in dimension $\leq 3$, \cite{TZ2016}. This can be thought of as an analogue of Perelman's convergence of the Ricci flow with surgery to a union of flat or hyperbolic pieces.

When Kodaira dimension satisfies $0<\kappa<n$, collapsing may occur. A typical example is the product $\Sigma\times T$ where $\Sigma$ is a higher genus curve and $T$ a torus. The normalized Ricci flow collapses the torus factor and the Gromov-Hausdorff limit is a constant curvature metric on $\Sigma$. More generally, one expects the fibers of a map $X\to X_{can}$ to the canonical model to collapse.

\begin{question}
Does the normalized Ricci flow converge in Gromov-Hausdorff sense to a generalized K\"ahler-Einstein space?
\end{question}

In \cite{TZ2018}, a relative volume estimate is used to control the size of fibers, and this allows to handle the case where $\kappa=1$.

\begin{thebibliography}{99}
\bibitem{TZ2016}
{\sc G. Tian, Zh. Zhang}
{\it Convergence of K\"ahler-Ricci flow on lower dimensional algebraic manifolds of general type},
 Inter. Math. Res. Not. {\bf 21} (2016), 6493--6511.
\bibitem{TZ2018}
{\sc G. Tian, Zh. Zhang}
{\it Relative volume comparison of Ricci Flow and its applications},
ArXiv: 1802.09506
\end{thebibliography}

\section{Totally geodesic submanifolds and positive curvature}
\proposer{Wolfgang Ziller, University of Pennsylvania}

Frankel's theorem, \cite{TF1961}, says that in a complete connected Riemannian $n$-manifold of positive sectional curvature, two closed totally geodesic submanifolds of dimension $n_1$ and $n_2$ must intersect provided $n_1 + n_2 \geq n$.

\begin{question}
Does Frankel's theorem hold for symmetric Finsler metrics?
\end{question}

We proved that the theorem holds in dimension two, but fails for non-symmetric Finsler metrics on $S^2$, \cite{BFIMZ}.

\begin{thebibliography}{99}
\bibitem{BFIMZ}
{\sc R.L. Bryant, P. Foulon, S. Ivanov, V. S. Matveev, W. Ziller},
{\it Geodesic behavior for Finsler metrics of constant positive flag curvature on $S^2$,}
arXiv:1710.03736
\bibitem{TF1961}
{\sc T. Frankel},
{\it Manifolds with positive curvature},
Pacific J. Math. {\bf 11} (1961) 165--174.
\end{thebibliography}

\section{Closed geodesics}
\proposer{Wolfgang Ziller, University of Pennsylvania, and Miguel Angel Javaloyes Victoria, Murcia}

It was recently proved that for a bumpy Finsler metric on $S^2$ (i.e., the energy function on the free loop space is Morse-Bott) there are either two or infinitely many closed geodesics, \cite{CGHP}.

\begin{question}
It is true without the bumpy assumption?
\end{question}

\begin{thebibliography}{99}


\bibitem{CGHP}
{\sc D. Cristofaro-Gardiner, M. Hutchings, D. Pomerleano},
{\it Torsion contact forms in three dimensions have
two or infinitely many Reeb orbits,}
arXiv:1701.02262
\end{thebibliography}

Frank Morgan

Williams College, Williamstown

\medskip

Pierre Pansu

Laboratoire de Math\'ematiques d'Orsay, Univ. Paris-Sud, CNRS, Universit\'e Paris-Saclay 91405 Orsay (France)


\end{document}
